

Article Robust Ship Detection in Infrared Images through Multiscale Feature Extraction and Lightweight CNN

Rui Miao ⁽⁾, Hongxu Jiang * and Fangzheng Tian

Beijing Key Laboratory of Digital Media, School of Computer Science and Engineering, Beihang University, Beijing 100191, China; miaor@buaa.edu.cn (R.M.); amazingtian@buaa.edu.cn (F.T.) * Correspondence: jianghx@buaa.edu.cn

Abstract: The sophistication of ship detection technology in remote sensing images is insufficient, the detection results differ substantially from the practical requirements, mainly reflected in the inadequate support for the differentiated application of multi-scene, multi-resolution and multitype target ships. To overcome these challenges, a ship detection method based on multiscale feature extraction and lightweight CNN is proposed. Firstly, the candidate-region extraction method, based on a multiscale model, can cover the potential targets under different backgrounds accurately. Secondly, the multiple feature fusion method is employed to achieve ship classification, in which, Fourier global spectrum features are applied to discriminate between targets and simple interference, and the targets in complex interference scenarios are further distinguished by using lightweight CNN. Thirdly, the cascade classifier training algorithm and an improved non-maximum suppression method are used to minimise the classification error rate and maximise generalisation, which can achieve final-target confirmation. Experimental results validate our method, showing that it significantly outperforms the available alternatives, reducing the model size by up to 2.17 times while improving detection performance be improved by up to 5.5% in multi-interference scenarios. Furthermore, the robustness ability was verified by three indicators, among which the F-measure score and true-falsepositive rate can increase by up to 5.8% and 4.7% respectively, while the mean error rate can decrease by up to 38.2%.

Keywords: ship detection; multiscale feature extraction; lightweight CNN; robustness

1. Introduction

Ship detection from infrared remote sensing images has an important but challenging role in remote surveillance and military reconnaissance [1,2]. Due to the large coverage area of infrared remote sensing images, and the small proportion of targets in the images, the accuracy of target-detection algorithms and real-time processing performances is seriously affected. For example, a remote sensing image obtained by one satellite contains $30,000 \times 30,000$ pixels, among which the size of ship area is 10×10 pixels, and the target area only accounts for one part per million of the image, which results in a serious lack of target texture detail, especially when the image resolution is low. Although certain existing infrared-image ship detection technologies have emerged, it is difficult for these methods to simultaneously address the following challenges: (1) it is difficult to extract universal features for various types of target ships in low-resolution or low-contrast images. Detection in low-resolution or low-contrast images will lead to the loss of texture details, and difficulty in the extraction of the available features of the target. Coupled with the interference of infrared image noise from clouds, reefs and so on, the features of the middle and lower layers are easily ignored, and it is easy to generate false-positives and negatives. (2) It is harder to maintain the balance between algorithm performance and algorithm complexity. Generally speaking, algorithms with high detection performances have high levels of complexity due to their calculation modes. When algorithms with high



Citation: Miao, R.; Jiang, H.; Tian, F. Robust Ship Detection in Infrared Images through Multiscale Feature Extraction and Lightweight CNN. *Sensors* 2022, *22*, 1226. https:// doi.org/10.3390/s22031226

Academic Editor: Federico Angelini

Received: 4 January 2022 Accepted: 2 February 2022 Published: 6 February 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). complexities need to be transplanted to embedded chips with limited resources, part of the algorithm's performance is often sacrificed. (3) Most of the existing detection methods compromise on real-time computing. The majority of detection algorithms have requirements that make them difficult to match with the resources that are available in real-time space reconnaissance applications, especially when introducing algorithms with the depth of neural networks to target classification; it not only requires a large amount of training data to be effectively generalized but also uses a larger amount of computational power compared to other methods.

Infrared images have a strong spatial correlation, contain more homogeneous regions, and have weak texture features, so the mean gray value is relatively stable [3]. However, when detecting a target ship on the sea surface, there are still several forms of interference: first of all, because the response characteristics of each pixel in the infrared imaging equipment are not completely consistent, different detection units produce different outputs under the same radiation input, so there is a bright and dark striped noise in the infrared image, resulting in a low signal-to-noise ratio of the image, which seriously affects the performance of the target-detection algorithm. Secondly, remote sensing satellites are easily affected by clouds, sea-clutter, and other weather during imaging, resulting in a complex detection background, reduced contrast between the target and background, which makes it easy to produce a large number of false-alarm. Thirdly, it is difficult to select appropriate features to separate the target and background because of the existence of various types and sizes of targets, and the unequal representation of grayscale features. Therefore, at present, there is no detection algorithm that is applicable to all possible scenarios, exist methods can only be used to minimise interference and ensuring detection efficiency in specific detection scenarios.

With the improvement of artificial intelligence technology, deep learning can adaptively and automatically learn features in data by constructing a deep neural network, which makes up for the deficiency of manual design features to a certain extent [4–6]. Object detection technology based on deep learning can generally be divided into one-stage and two-stage detection methods. Two-stage target-detection algorithms generally consist of candidate-region extraction and target confirmation, they utilize powerful features with statistical classifiers to discriminate ships from false-alarm, which has great advantages in maintaining detection accuracy. One-stage target-detection algorithms aim to extract all the candidate regions for a subsequent classifier, they omit the step of candidate-region extraction and directly obtain the target category and position from the image. Compared with two-stage algorithms, they have a huge speed advantage, but their detection accuracies are low due to their rough detection strategy. When deep learning algorithms are deployed on embedded platforms or other platforms, it is a great challenge to balance the accuracy, speed and memory resources needed for target detection.

As a result of the above-mentioned analysis, the complex background and diverse interference factors will seriously affect the extraction and classification of the effective features of the targets. In addition, during the process of deployment and application, it is difficult to balance detection performance, computational complexity and real-time performance. In order to solve the problems raised above, the contributions of this paper are presented as follows. (1) Candidate-region extraction: our method combines the cascade rejection mechanism with multiple other features through a linear cascade classifier, which orders candidates from simple to complex, uses relatively simple features to exclude a large number of simple alarms, such as seawater and clouds, and uses more sophisticated features to extract final candidate regions. (2) Multiple feature fusion-classification: a falsealarm elimination method based on Fourier global spectral features and lightweight CNN is proposed. In this method, the global Fourier transform was applied to each candidate area to obtain the corresponding feature description and achieve a rough classification of candidate regions, then the local feature was extracted by the lightweight CNN model to further eliminate false-alarm. (3) Classifier training and target confirmation: a classifier training algorithm is proposed to minimise the classification error rate and maximise

3 of 24

generalisation, and the improved NMS algorithm is used to merge real ships and achieve an accurate output.

The remainder of this paper is organized as follows. Section 2 describes the related work. Section 3 introduces the methodology and details the elaborate implementation and optimization of the core module of the algorithm. Section 4 describes the extensive experiments, and Section 5 presents our conclusions and recommendations future work.

2. Related Work

In recent years, many infrared ship detection algorithms have been proposed by researchers. In these researches, ship target detection algorithms are generally divided into traditional ship detection method and deep learning ship detection method. The existing traditional ship detection algorithms can be divided into four categories for different scenarios: ship detection algorithms based on wake extraction; ship detection algorithms based on template matching; ship detection algorithm based on feature statistics; and ship detection methods based on classification learning. Although some scholars have put forward some novel research ideas, their core ideas are inseparable from the above types of target detection. Traditional target-feature extraction methods are mainly based on the idea of grey and texture features where a pre-trained classifier is employed for classification. For example, some researchers propose a saliency strategy, a feature descriptor [7,8], and a local comparison method [9] to determine small infrared targets, but these methods are very sensitive to noise, which usually generates a high false-alarm rate. Most infrared small-target-detection methods based on saliency have high computational complexity and are difficult to optimise using parallelism. Therefore, ship detection algorithms based on the weighted local difference measurement [10], and weighted voting mechanism [11] have been proposed. Moreover, references [12,13] introduce the multiscale local uniformity and greyscale difference weighting strategy to detect small infrared targets. In references [14,15], multi-frame images and sensor data in the infrared image sequence were analysed for ship detection. Based on extreme value theory, the edge detection [16,17] and cascading characteristics methods [18] are adopted to identify the objects of interest and suppress background clutter. The above traditional detection methods are often based on low-level, hand-made features. It is a great challenge to achieve high detection accuracy in complex scenes, such as those with cloud interference and low contrast, and improvement is needed.

With the improvement in artificial intelligence technology, deep learning has attracted an increasing amount of attention. Target-detection techniques based on deep learning include anchor-based and anchor-free techniques. Firstly, anchor-based technology includes one-stage and two-stage detection. One-stage detection techniques include the single-shot detector (SSD) [19], the deconvolutional SSD (DSSD) [20], RetinaNet [21], RefineDet [22], You Only Look Once Version 3 (YOLOV3) [23], etc. Two-stage detection techniques include FTP-region-based convolutional neural networks (RCNNs) [24], region-based fully convolutional networks (R-FCNs) [25], the Feature Pyramid Network (FPN) [26], Cascade R-CNNs [27], the subnet Internet Protocol (SNIP) [28], etc. Generally, two-stage target detection is more accurate than one-stage target detection, but the processing speed is slow. Secondly, anchor-free technology includes key-points and segmentation. Key-pointbased technologies include CornerNet [29], CenterNet [30], Cornernet-Lite [31], etc., and segmentation technologies include a feature-selective anchor-free (FSAF) module [32], a fully convolutional one-stage (FCOS) object detector [33], FoveaBox [34], etc. These deep learning methods achieve good detection accuracy in natural image target detection, but they also have great limitations during satellite processing with limited remote sensing image resources. First, the compression method of deep neural networks has higher performance requirements, especially for large networks. Second, the algorithm has difficulty meeting performance expectations. It is difficult to design a state-of-the-art machine for the data-flow scheduling of different layers, and there will be considerable redundancy in logical resources. In addition, there is the data dependence problem. Compared with traditional methods, deep learning relies more on the large-scale training of data and it

needs a large amount of data to understand the potential data-mode. When the target features and false-alarm features in the detected images are relatively uniform, such a data dependence problem is not obvious, otherwise, when the target features and false-alarm features in the detected images are significantly different, the scale of training data will need to be considerably increased, which is a great practical challenge.

In some relatively simple conditions, the methods mentioned above can achieve considerable detection results. However, the detection performance of these algorithms will be affected in the following three situations: (a) low contrast between ships and background; (b) scenes with complicated sea conditions; and (c) in situations of false-alarm interference. In addition, these algorithms also give rise to different levels of missed detection when multiple vessels are docked. Therefore, there is still much room for improvement in ship detection.

3. Methodology

3.1. Candidate-Region Extraction

To solve the problem of complete extraction of the ship region, the suspected region of the target ship is located step-by-step based on the idea of coarse-to-fine detection, as is shown in Figure 1. We design the candidate-region extraction through three parts. Firstly, using a multi-scale model, hierarchical images are constructed, which are employed as original image-data for subsequent processing. Secondly, regional gradient-feature reconstruction is undertaken. The region of interest (ROI) is extracted by constructing the Sobel operator and gradient template, and the target region is preliminarily determined. Thirdly, in the vector binarization of flow convolution, target information is extracted from the combination of multiple features by traversing the image data using flow convolution alongside using the more sophisticated features for further identification.



Figure 1. Schematic of the candidate-region extraction process.

3.1.1. Multiscale Model

The target types to be detected are different under different image resolutions, such as in 5-m-resolution images, the target ships are generally large military targets, large passenger targets and large cargo targets, but in high-resolution images, fishing ships and other small ships are also the targets that need to be detected. Ships in the same remote sensing image have different sizes, and the same ship has different scales in images with different resolutions. Therefore, the existing methods are difficult to adapt to problems that involve large differences in the apparent features of ship caused by scale diversity. In order to find target ships of different sizes in remote sensing images, we constructed a standard image model through repeated smoothing and sub-sampling, and then generated a multi-scale image model through reasonable scaling. The definition of each layer is shown in Equation (1).

$$Lay_{i} = Lay_{ori} \otimes \theta_{i}, i = 1, 2, 3...n$$
(1)

where Lay_i represents the generated image models at different scales, Lay_{ori} represents the standard image model, \otimes represents the bilinear interpolation operation, θ_i represents the scaling factor, and the value of parameter i ranges from 1 to n. In the candidate-region extraction stage, one of the more important parameters is the filter size (k × k), which is the sliding cell that we need. As to the small target, we can set the filter size to ensure that it completely covers the target. As for the larger ships, if a smooth or partial calculation can't cover the whole target, we down-sample the image according to the actual testing requirements in terms of narrowness, until the scaled target size meets the minimum coverage area. Therefore, the selection of the parameter "k" is very important. If k is too small, the target ship cannot be covered, otherwise, false-alarm will appear in the covering box. By analyzing the size of target ship in the dataset, we can select parameters 10–15, basically to detect all ships in the dataset on the basis of a multi-scale model, this size-range is the optimal processing unit size after a lot of derivation, and can deal with different target sizes and characteristics. In this paper, we define k as 11, that is, the filter size is 11 × 11 pixels.

The parameter θ relates to the reduction factor for image scaling. For easy understanding, the parameter θ can be regarded as the enlargement factor for the sliding 11×11 window. In the multi-scale model stage, there are two important parameters: the image size img \times img; and the threshold value θ . Since the size of the sliding window is 11 \times 11, the parameters should be set to fully cover the target when using the maximum reduction factor, where the maximum reduction factor is the result of the multiplication of θ_1 , θ_2 and θ_3 . Through a large amount of training and derivation, we find that when the scale is too small it will result in incomplete coverage, and if the scale is too large it will result in the loss of target features. When the maximum reduction factor is close to 35, better results can be obtained in the data set. Further tests of the combination of factors in each layer achieve combinations of values of 2, 1.25 and 1.25 for the maximum reduction factor. Under this set of parameters, our algorithm can achieve a better detection effect. Of course, this specific combination is not unique, we only need an approximate optimal combination, and the parameters θ_1 , θ_2 and θ_3 can be exchanged freely. Therefore, in order to meet the test requirements, if we make $\theta_1 = 2$, $\theta_2 = 1.25$, $\theta_3 = 1.25$, we design a three-layer scale scaling model that can meet the full coverage requirements of all ships of different sizes, and the maximum coverage size can reach 34 pixels, which meets the test requirements of ship samples in the dataset. At the same time, because the multi-scale model uses a standard image zooming process for large object detection without any processing of the standard image for testing, it can also provide security for small-target detection (e.g., pixels covering far less than 11×11 of a small target in a standard image can be detected even if the target is not detected after scaling through the entire range of scales in the model), and avoid the occurrence of missed detection problems.

Since the sizes of images obtained in the detection stage are different, to facilitate subsequent processing, the first stage of this paper is to carry out image scaling. For vector images, image scaling does not cause distortion, blur or other problems, but the

infrared remote sensing image is similar to a bitmap, and thus, it is necessary to select an appropriate image scaling algorithm. Bilinear interpolation is the interpolation of image pixels. Even under the condition that the original image is not smooth, bilinear interpolation will produce a smooth output, and as the infrared remote sensing image resolution is low, the target ships in the images require contour smoothing if they are small. In this case, the bilinear interpolation algorithm has a high-quality effect; the algorithm complexity is lower, and the time efficiency is better.

For point P = (x, y) on the line between $Q_1 = (x_0, y_0)$ and $Q_2 = (x_1, y_1)$, the calculation process of the y coordinate is shown in Equation (2):

$$y = y_0 + (x - x_0) \frac{y_1 - y_0}{x_1 - x_0} = \frac{y_0(x_1 - x) + y_1(x - x_0)}{x_1 - x_0},$$
(2)

From the perspective of the weighted average, the weight is inversely proportional to the distance between the known point and the unknown point, that is, the closer the known point is to the unknown point, the greater the weight of the solution result of the unknown point. Therefore, according to the distance between the normalised unknown point and the two known points along the *X*-axis, the two weights should be $\frac{x-x_0}{x_1-x_0}$ and $\frac{x_1-x}{x_1-x_0}$, respectively. We obtain the derivative calculation progress of y that is shown in Equation (3) as follows:

$$y = y_0 \left(1 - \frac{x - x_0}{x_1 - x_0} \right) + y_1 \left(1 - \frac{x_1 - x}{x_1 - x_0} \right) = y_0 \left(1 - \frac{x - x_0}{x_1 - x_0} \right) + y_1 \left(\frac{x - x_0}{x_1 - x_0} \right), \quad (3)$$

Bilinear interpolation is the extension of linear interpolation in the plane region, and its core idea is to perform linear interpolation in each direction of two dimensions. Given the coordinates of four points $Q_{11} = (x_1, y_1)$, $Q_{21} = (x_2, y_1)$, $Q_{12} = (x_1, y_2)$, and $Q_{22} = (x_2, y_2)$, and given the value of function f at four points, the bilinear interpolation algorithm can be utilised to obtain the value of function f at point P(x, y). The calculation process is shown in Equations (4)–(6):

$$f(x, y_1) = \frac{x_2 - x}{x_2 - x_1} f(Q_{11}) + \frac{x - x_1}{x_2 - x_1} f(Q_{21}),$$
(4)

$$f(x, y_2) = \frac{x_2 - x}{x_2 - x_1} f(Q_{12}) + \frac{x - x_1}{x_2 - x_1} f(Q_{22}),$$
(5)

$$f(x,y) = \frac{y_2 - y}{y_2 - y_1} f(x,y_1) + \frac{y - y_1}{y_2 - y_1} f(x,y_2),$$
(6)

In this paper, a bilinear interpolation algorithm is used to expand and shrink the image size of the module to be detected, and images of different sizes are obtained as the input of the subsequent detection module. Through this scaling process, the subsequent detection algorithm has better adaptability to target ships with different characteristics. A summary of the multiscale model's operation is shown in Algorithm 1.

3.1.2. Regional Gradient-Feature Reconstruction

The target in the image usually has a well-defined contour, while most backgrounds do not. According to the difference between the target and the background, the background and target can be distinguished by gradient features.

To quantify the possibility of targets being contained in a region, the region size is adjusted to a fixed size, and then the gradient of the whole region is calculated as the feature vector. For image I, I (i, j) represents the grey value at position (i, j) in the image. The gradient x_i of image I along the x-direction is the first derivative of the image in the x-direction. The gradient y_i of image Img along the y-direction is the first derivative of the

image in the y-direction. The calculation of gradients I_x and I_y is shown in Equations (7) and (8), respectively:

$$I_x = \frac{I(i+h,j) - I(h,j)}{h}$$
(7)

$$I_y = \frac{I(i,h+j) - I(h,j)}{h}$$
(8)

In the actual algorithm design process, the value of h is generally 1, and the Sobel operator is applied to quickly extract image gradient features. Sobel operator is a discrete difference operator, which is used to approximate the gray-level from an image brightness function. Using this operator at any point in the image will produce the corresponding grayscale vector. The Sobel operator is based on the gray-weighted-difference between the upper, lower, left, and right adjacent points of a pixel. If the gradient value of a pixel in the overall x and y directions is obtained, it only needs to add the gradient calculation results calculated by Sobel in each direction. The Sobel operator is used to calculate the gradient characteristics of the image, which is shown in Equations (9) and (10) as follows:

$$I_x = S_x \otimes Img \tag{9}$$

$$I_{v} = S_{v} \otimes Img \tag{10}$$

where Img represents the input image, S_x and S_y represent the Sobel operators, and \otimes represents the gradient operation. The gradient amplitude M (i, j) of image Img at (x, y) can be obtained from I_x and I_y , which are shown in Equation (11):

$$\mathbf{M}(\mathbf{i},\mathbf{j}) = \min(|\mathbf{I}_{\mathbf{x}}(\mathbf{i},\mathbf{j})| + |\mathbf{I}_{\mathbf{v}}(\mathbf{i},\mathbf{j})|, \mathbf{S}_{\text{image}})$$
(11)

Algorithm 1 Multiscale model operation

Input: Blocks of images, location to be processed, down-sampling parameter θ Output: "ResamImg" represents the output layers after down-sampling 1. Do the following steps: 2. Initialise parameters of first layer θ = 2, initialisation Loc.x, Loc.y, Loc.height, Loc.width, and ResamImgylen; 3. Update ResamImg as following condition: for lav = 1pos = detector (ResamImg, model, Opts); $Loc.x = [Loc.x, pos.x \times scale + xstart];$ Loc.y = [Loc.y, pos. y \times scale + ystart]; Loc.height = [Loc.height, pos. height \times scale]; Loc.width = [Loc.width, pos. width × scale]; ResamImg_xlen = round(ResamImg_xlen/Opts.sStep); ResamImg_ylen = round(ResamImg_ylen/Opts.sStep); ResamImg = imresize (ResamImg, [ResamImgylen, ResamImg xlen], 'bilinear'); scale = 2;end 4. End

Figure 2 shows a schematic of the image after scaling. The target ship can be better represented by selecting an appropriate scaling factor.

3.1.3. Vector Binarization of Flow Convolution

The operation of linear convolution is essentially the computation of the inner product of two vectors by representing a vector as the weighted sum of multiple binary vectors, consisting of -1 and 1, such that the inner product operation of a vector can be computed quickly using simple bit operations. The convolution operation is an important step in the realisation of whole-target detection. When hardware is employed to realise the convolution operation, it is necessary to identify a scheme that accurately realises the convolution operation. The operation does not occupy too much space and is fast enough to complete the convolution operation of a target region in a pixel-clock. During the convolution operation, it is necessary to know the value of pixels around the current pixels. However, in the process of image processing, the data obtained exhibit the form of pixel streams rather than the whole image, and thus, it is necessary to cache the surrounding pixels that are needed. Algorithm 2 shows a summary of the vector binarisation operation.



Figure 2. Diagram of image gradient operation: (**a**) Original image with target ship, (**b**) image with target after gradient operation, (**c**) original image with background, and (**d**) image with background after gradient operation.

Algorithm 2 Vector binarisation operation
Input: Vector w to be approximated, number of binary vectors N_w
Output: $\{\{\partial_j\}_{j=1}^{N_w}\}, \{\{\beta_j\}_{j=1}^{N_w}\}$, represents N _w binary vectors and corresponding weights
1. Do the following steps:
2. Initialise residuals $\varepsilon = w$
3. Update ∂_i and β_i as the following conditions:
for $j = 1$ to N_w do
$\partial_{j} = \operatorname{sign}(\varepsilon)$
$\beta_j = \langle \partial_j, \varepsilon \rangle / \partial_j ^2$
$\varepsilon \leftarrow \varepsilon - \beta_j \alpha_j$
end
4. End

The approximate representation of w can be obtained through the vector binarisation approximation algorithm, $w \approx \sum_{j=1}^{N_w} \beta_j \partial_j$, assuming that the region size is 8 × 8. The gradient feature is a 64-dimensional feature vector, $x \in R^{64}$. The model parameters of the linear support-vector-machine (SVM) classifier based on gradient-feature training are equivalent to the gradient feature dimension, which is also a 64-dimensional vector, and is defined as $w \in R^{64}$. In the above summary of Algorithm 2, ∂_j represents the binary basis vector $\partial_j \in \{-1, 1\}^{64}$. After the binary approximation algorithm, there are a total of N_w binary basis vectors, and β_j represents the weight coefficient corresponding to the basis vector ∂_j .

The pixels around the target pixels are stored in registers according to their addresses. During the algorithm's operation, values flow from left to right and top to bottom throughout the image as required. The next step is to convolve the pixel value stored in the register with the corresponding convolution-kernel weight. The ping-pong operation is a commonly used data-flow control-processing technique, the main process is to assign input data streams to different data buffers isochronously through an input-data selection unit. The ping-pong operation sends the buffered data-stream to the processing module continuously for calculation by an input-data selection unit and an output-data selection unit, switching with each other according to the beat. Because the input-data- and output-data-flows are continuous, it is very suitable for the pipeline processing of data-flows to complete seamless buffering and processing of data, greatly saving buffer space. In this paper, the number of convolution kernels is constant, a new pixel can be input in a pixel-clock-cycle, and the convolution result can be output after several clock-cycle delays. If the ping-pong operation is not used, the data preprocessing module will become the bottleneck to limit the system-data-throughput in the design. By optimizing the ping-pong operation's design, the computation period is increased. The data throughput of the system can be improved through the cache design optimisation while increasing the data buffer delay. This process can be defined as follows:

$$Conv = Max(conv(T_{xn'}C_{xn})$$
(12)

where T_{xn} represents the pixel matrix, which is processed according to the regional gradient, C_{xn} represents the convolution template, the templates in this article are 11 × 11 pixels, and there are six template types in different directions, as shown in Figure 3. Max represents the maximum pixel matrix after convolution of the image matrix and several pixel matrices.



Figure 3. Diagram of six templates with different directions in the convolution module.

The value of each pixel of Conv is compared with the detection threshold ξ . A value greater than ξ is regarded as a suspected target ship, and the coordinate information of the target is output. Otherwise, it is regarded as a nontarget point. To choose an appropriate value of η , the recall rates are counted with different choices of ξ , which are shown in Table 1. We can see that a value of $\xi = 0.6$ is a turning point in the recall rate. When ξ is set larger than 0.6, the recall rate quickly decreases. On the other hand, when ξ is set smaller than 0.6, it will slow down the subsequent processes and bring additional false-positives. So, we choose $\eta = 0.6$ in our method as this is the optimal parameter-solution proved by a large number of tests. In addition to ensuring that the recall rate of the algorithm remains within an appropriate range, the real-time processing performance of the algorithm is maximally improved. The target information output by the threshold model is regarded as the input for region classification, in which the target information mainly includes an x-coordinate, a y-coordinate, a width and a height.

Table 1. The effect of threshold on algorithm performance.

ξ	0	0.2	0.4	0.6	0.8	1
Recall rate	1	0.999	0.996	0.992	0.756	0
Time/img	1.998	1.996	1.986	1.962	1.42	0

3.2. Multiple-Feature Fusion Classification

To solve the problem of complete extraction of the ship region, the suspected region of the target ship is located step-by-step based on the idea of coarse-to-fine detection, as is shown in Figure 4.

We design the multiple-feature classification process through two parts. The first is Fourier global spectral feature extraction. The global spectral features based on Fourier transform are applied to extract the Fourier features of positive and negative samples to train the classifier. The magnitude of the gradient generated by the Fourier global spectral feature represents the difference between a point in the image and its neighbourhood, which can better distinguish the target ship from the ocean background and initially exclude false-alarm. Secondly, local feature classification through lightweight CNN is undertaken. Through the full analysis of target ships in infrared remote sensing images, available features that can effectively distinguish target ships from typical false-alarm (clouds, tracks, etc.) are selected. The optimal feature-subset of the target can be constructed to quickly and accurately eliminate false-alarm to improve the accuracy and universality of ship detection.



Figure 4. Schematic of the multiple-feature fusion classification process.

3.2.1. Fourier Global Spectral-Feature Extraction

Through the function of frequency domain analysis, we can change the angle and improve the visibility of certain signal information. For example, by Fourier transform [35,36], if the numerical value is graphed and existing spikes are visible, they not only represent hidden frequencies in the mixed signal but also can solve the problem of distinguishing between several signals. Specifically, the work of this stage is to perform Fourier transform on the candidate slices obtained in the previous stage. For each candidate target image, the spatial domain is converted to the frequency domain. In other words, a two-dimensional Fourier transform is applied to each candidate region to obtain the spectral domain, which is the distribution of the image gradient. The magnitude of the gradient represents the strength of the difference between a certain point on the image and the neighbourhood point, which is used to better distinguish the target ship from the ocean background. During the training, the Fourier transform is extracted from the positive sample set of six different directions (ship direction up, down, left, right, left tilt 45 degrees, and right tilt 45 degrees) to extract the spectrum characteristics. The diagram of six templates is shown in Figure 5.





Figure 5. Diagram of six templates with different directions in Fourier module.

More specifically, each sample slice is extracted from six parts of theimage block, among which, the first part is the whole area of the square data block, the second part is the upper part of the square data block (width 32, height 8), the third part is the left part of the square data block (width 8, height 32), the fourth part is the lower part of the square data block (width: 32, height: 8), the fifth part is the right part of the square data block (width: 8, height: 32), and the sixth part is the middle part of the square data block (width: 16). These specific extracted features are shown in Figure 6.



Figure 6. Diagram of six parts of mage block. From left to right is the global part, the upper part, the left part, the right part, the lower part, and the middle part.

To be more specific, each sample section is extracted from the spectral features of one global region, four background regions, and one central region to which the target belongs. The two-dimensional Fourier transform is defined as follows:

$$F(u,v) = \sum_{x=0}^{M-1} \sum_{y=0}^{N-1} f(x,y) e^{-j2\pi (\frac{ux}{M} + \frac{vy}{N})}$$
(13)

where u and v are frequency variables, the value of u ranges from 0 to M - 1, and the value of v ranges from 0 to N - 1. F(x, y) represents the graph function, and M, N represents the length of the sequence f(x, y). The feature descriptions of the entire global slice I_1 , the centre region I_2 containing the target ship, and the background region I_3 except the centre are f_1 , f_2 , and f_3 . The calculation process refers to Equation (14) as follows:

$$\mathbf{f}_{i} = \mathbf{A}(\boldsymbol{\zeta}[\mathbf{I}_{i}(\mathbf{x})]), \tag{14}$$

Among them, the value of i ranges from 0 to 5, and A is the Fourier transform operation. The frequency-domain descriptor F is obtained by fusing the global and local features of the image, $F = [f_0; f_1; f_2; f_3; f_4; f_5]$. Similarly, we perform a similar task for a negative sample set. The spectrum feature vectors obtained by the Fourier transform of the positive and negative sample sets in the abovementioned steps are sent to a classifier. In the experimental process, Fourier global features are applied to roughly eliminate false-alarm and extract several candidate areas. For easy understanding, the six-part solution of the Fourier transform is shown in Algorithm 3.

The global Fourier transform was applied to each candidate frame to obtain the corresponding feature description, achieve a rough classification of candidate regions, and then the local feature was extracted by the lightweight CNN model to further eliminate false-alarm in the following section.

Algorithm 3 Fourier operationInput: Original image (img) to be approximated.Output: Final vector feature, number of binary vector N_f .1. Do the following steps:2. Initialise residual feature (i)3. Update feature (i) as the following conditions:im = im2double(im);IM = abs(fft2(im)); feat0 = IM (:)'; feat0(1) = 0;IM = abs (fft2(im (1:10:))); feat1 = IM (:)'; feat1(1) = 0;IM = abs (fft2(im (:1:10))); feat2 = IM (:)'; feat2(1) = 0;IM = abs (fft2(im (end-7:end:))); feat3 = IM (:)'; feat3(1) = 0;IM = abs (fft2(im (end-7:end:))); feat4 = IM (:)'; feat4(1) = 0;IM = abs (fft2(im (end/4+1:end/4×3, end/4+1:end/4×3))); feat5 = IM (:)'; feat5(1) = 0;feat = [feat0, feat1, feat2, feat3, feat4, feat5];feat = feat./(sum(feat));

4. End

3.2.2. Local Feature Classification through Lightweight CNN

Generally, when deep learning methods are utilised to extract features for classification, deeper network layers and larger datasets are needed to achieve higher classification performance. However, the larger the model, the larger the number of parameters and computing resources that are consumed. It is difficult to meet the limited resource requirements of the onboard processor. Therefore, to balance accuracy, speed and memory, the global features of positive and negative samples are extracted with the Fourier spectrum features in the previous part to roughly eliminate false-alarm in the candidate region of the target ship. A lightweight classification network is then designed to accurately identify the local features of the ship. The reason for combining the Fourier global features with local features extracted from lightweight networks in this chapter is to consider the following two aspects. (a) In general, the full connection layer in deep networks learns global patterns from the feature space, and the convolution layer learns local patterns, while lightweight networks can only learn simple features at a lower level. (b) With an increase in the network layer number, the receptive fields are also increased gradually. If the network layer is low, the convolution kernels are lower, and the receptive fields are unable to capture the global image. Therefore, the Fourier spectrum of global features can effectively reduce the size of the previous stage to produce the number of invalid candidate areas, reducing the complexity for the lightweight classification-network classification target.

The target ship itself in the external remote sensing image has a small scale of approximately 10 to 50 pixels. Considering the small scale of the target ship itself, the model compression will compensate for the loss of accuracy. Therefore, lightweight networks with fewer network layers are designed in this chapter. The lightweight classification network consists of four convolutional layers and two fully connected layers. The network structure is shown in Figure 7. Among them, the first three convolutional layers are configured with a convolution kernel with a size of 3×3 , followed by the maximum pooling layer of downsampling. The last convolution layer is configured as a 1×1 convolution kernel. Deep convolution and 1×1 convolution can reduce the computational complexity of the model with a small precision loss. These convolutional layers are deployed on rectified-linear activation-unit (ReLU) activation functions to make the network undisturbed by vanishing gradients compared to sigmoID activation functions and tanh activation functions. In addition, different data enhancement strategies are adopted in the training process. The last fully connected and softmax layers address the classification problem based on the features extracted from the previous convolutional layer. During the experiment, cross-entropy was employed to define the loss function as follows:

$$Loss(Y, P) = -\frac{1}{N} \sum_{i=0}^{N-1} \sum_{K=1}^{2} (y_{ik} \log(p_{ik})),$$
(15)

where N is the batch size, which is 128 in this experiment. The term i represents the label of the candidate slice, y_{ik} represents that the candidate slice label of block I belongs to category K, and p_{ik} represents the predicted probability that the candidate slice belongs to category K. In order to facilitate understanding, the extraction and calculation of feature vectors at each layer are briefly shown in Algorithm 4.

Algorithm 4 Local feature extraction

Input: N training pictures of size $m \times n$,

Output: Parameters and thresholds for each classifier comprise the output

1. Do the following steps:

2. Initialise and normalize training image size and the number of filters;

3. N training pictures can be obtained: $I = [\bar{I}_1, \bar{I}_2, \bar{I}_3...\bar{I}_n]$, where I is an image obtained after rearrangement and preprocessing of each image.

4. Filter at every stage can be expressed as: $\psi_f^1 = e_f(\mu^T)$, $f = 1, 2, 3...L_1$, where $e_f(\mu^T)$ denotes the f principal eigenvector of μ^T ;

5. Each obtained image was preprocessed, and the results of image segmentation were merged together, compute the block result of N pictures and one of the filter convolutions $\phi^i = [\phi^1, \phi^2 \dots \phi^{L_1}]$

6. By solving the eigenvector of $\varphi \varphi^T$, the feature vector corresponding to the second largest eigenvalues was taken as a filter, $\psi_f^2 = e_f(\varphi \varphi^T)$, $f = 1, 2, 3...L_2$;

7. Similarly, we can calculate: $\psi_f^3 = e_f(\phi\phi^T)$, $f = 1, 2, 3...L_3$, $\psi_f^4 = e_f(\phi\phi^T)$, $f = 1, 2, 3...L_4$

8. Through spatial pyramid pooling, hash coding, etc., the feature vector of each training image was obtained.

9. Input the trained feature vectors into the LibSVM to train and test them





Figure 7. Schematic of feature extraction through a lightweight CNN.

This section proposes a lightweight convolutional network based on the target candidate region under resource constraints in real scenes. Based on the multi-scale target candidate-region extraction model, the network orders the candidate region to identify the real target and false-alarm, returns the index corresponding to the correctly classified candidate region to the corresponding connected region in the original image, and realises the target location according to the largest connected region's outer rectangle. In this way, the regressive operation of object detection based on deep learning is avoided, and the amount of computation is doubled. Therefore, the computational complexity of the model with a small precision loss can be reduced.

3.3. Classifier Training and Target Confirmation

As a powerful classification method with the ability to minimise the classification error rate and maximise generalisation, the basic working principle of the SVM is described as follows [37]: two kinds of samples that are linearly indivisible in the input space are mapped to a high-dimensional feature space by a kernel function, and linearly constrained quadratic programming is solved in the high-dimensional feature space to obtain a classification hyperplane with a maximum interval that can linearly divide the samples.

For the dataset $T = \{(x_1, y_1), (x_2, y_2), \dots, (x_m, y_m)\}, yi \in \{-1, +1\}$ and the hyperplane (w, b), the geometric interval of sample points (x_i, y_i) is defined as $\gamma_i = y_i(w/||w|| \cdot x_i + b/||w||)$, and the minimum geometric interval of the hyperplane (w, b) with respect to the training dataset T is $\gamma = \min_{i=1,2,\dots,N} \gamma_i$. Solving a separated hyperplane with the maximum geometric interval requires the maximum and minimum geometric interval, which can be expressed as the constrained optimisation problem $\max_{w, b} p_{w, b}$.

Considering the relationship between the geometric interval and the function interval, linearly separable support vector machines can be transformed into an optimisation problem as follows:

$$\min_{w,b} \frac{1}{2} ||w||^2 + s.t.y_i(w \times x_i + b) - 1 \ge 0 (i = 1, 2, \dots, N)$$
(16)

For linear indivisibility, a soft interval can be considered to allow some samples to fail to meet constraint conditions; the optimisation objective function is then defined as follows:

$$\min_{w,b} \frac{1}{2} ||w||^2 + C_p \sum_{i=1}^n \xi_i, \text{ s.t.} y_i(w \times x_i + b) \ge 1 - \xi_i$$
(17)

where C represents the degree of punishment for the right and wrong samples, and the optimisation of the whole algorithm can still use the Lagrange multiplier method. It can be seen from the optimisation objective function that an SVM ultimately achieves a compromise between the maximum classification interval and the minimum classification error, and its punishment is the same for positive and negative classification errors. In the process of extracting potential regions of ships, the classifier must ensure a high recall rate to avoid missing detection as much as possible. Therefore, the risks brought about by classifying ships as backgrounds and classifying backgrounds as targets differ, and the misclassification of the two cannot be treated equally. According to this feature, this paper demonstrates how the optimisation objective function improves and that the constructed-risk unbalanced SVM classifier overcomes the shortcomings of the traditional SVM classifier. In addition, the risk-unbalanced SVM classifier is applied to the task of ship potential-extraction. The optimisation objective function of an SVM with uneven risk is shown as follows:

$$\min_{w,b} \frac{1}{2} ||w||^2 + C_p \sum_{y_i=1} \xi_i + C_n \sum_{y_i=-1} \xi_i, \text{ s.t. } y_i(w \times x_i + b) \ge 1 - \xi_i$$
(18)

where C_p and C_n are the risk of positive samples and the risk of negative samples, respectively, and generally $C_p > C_n$. The flow of the cascade classifier training algorithm is shown in Algorithm 5.

For the same target, the classification algorithm may identify several bounding boxes. One target corresponds to multiple bounding boxes, meaning that it is necessary to filter out the redundant windows and retain only the optimal bounding boxes. Non-maximum suppression searches for the local maximum value of all bounding boxes to identify the maximum value within a certain neighbourhood retains the windows with the highest score in the neighbourhood and inhibits windows with low scores to filter part of the bounding boxes to improve the final detection accuracy.

Non-maximum suppression is an iteration-ergo-elimination process; the specific algorithm is expressed in Algorithm 6.

The traditional non-maximum suppression (NMS) algorithm generates a series of detection candidate boxes B and the corresponding probability value S. First, we sort the probability values and select the candidate box M with the maximum probability before M joins the final detection result set D and deletes it from B. Candidate boxes in set B whose overlap with candidate box M is greater than the threshold Nt are deleted. The main problem of the algorithm is that the adjacent detection frames must be deleted. In this case, if ships are clustered and distributed, there is an overlap between two detection

frames, which causes the detection failure of ships near each other and reduces the detection accuracy of the algorithm.

Algorithm 5 Cascade classifier training algorithm

Input: Training set $T_1 = \{(x_1, y_1), (x_2, y_2), \dots, (x_{m1}, y_{m1})\}$, where $x_i \in X, y_i \in Y = \{-1, +1\}$, given the weights of the two classes C_p, C_n and the minimum recall rate d_{min} Output: Parameters and thresholds for each classifier comprise the output 1. Do the following steps: 2. Calculate the variance of each sample in training set $T_1 = \{(x_1, y_1), (x_2, y_2), \dots, (x_{m1}, y_{m1})\}$ 3. Through all positive samples in the training set, adjust the threshold to make the recall rate d_1 of the classifier meet $d_1 \ge d_{min}$ 4. Eliminate the training samples marked as background by variance classifier from the negative sample set of the training set $T_1 = \{(x_1, y_1), (x_2, y_2), \dots, (x_{m1}, y_{m1})\}$. The training set becomes $T_2 = \{(x_1, y_1), (x_2, y_2), \dots, (x_{m2}, y_{m2})\}$ 5. Calculate the gradient characteristics of each sample in the training set $T_2 = \{(x_1, y_1), (x_2, y_2), \dots, (x_{m2}, y_{m2})\}$ 6. Train a linear SVM classifier according to the gradient characteristics of positive and negative samples of the training set $T_2 = \{(x_1, y_1), (x_2, y_2), \dots, (x_{m2}, y_{m2})\}$ and C_p, C_n . Adjust the threshold to make the recall rate d_2 of the classifier meet the requirement $d_2 \ge d_{min}$ Output the parameters and classification threshold of the cascade classifier 8. End

Algorithm 6 Improved NMS algorithm

Input: Collection of bounding boxes $B = \{b_1, b_2, ..., b_n\}$, the score of the bounding box $S = \{s_1, s_2, ..., s_n\}$, and the threshold value N_t **Output:** D, S 1. **Do** the following steps: 2. Initialize $D = \{\}$ 3. While B is not empty compute the index of the maximum value in S: $m \leftarrow \operatorname{argmax} S$ Compute the corresponding bounding box: $M \leftarrow b_m$ 4. Update s, D, B and S as the following condition: $D \leftarrow D + M, B \leftarrow B - M$ for b_i in B, do if iou(b_i, M) is present, update $S \leftarrow S + s_i, B \leftarrow B - b_i$ end 5. **End**

To solve this problem in NMS, this paper adopts a probabilistic reset strategy to improve NMS [38]. In this algorithm, the probability of an overlapping detection frame is directly obtained by an attenuation function rather than zero. If a detection frame overlaps greatly with M, it has a very low probability of still existing. In contrast, if the detection frame only overlaps to a small extent with M, its detection probability is not affected. Its probability reset function is shown as follows:

$$S_i = \begin{cases} S_i & \text{iou}(M, b_i) < N_t \\ S_i(1 - \text{iou}(M, b_i)) & \text{iou}(M, b_i) \ge N_t \end{cases}$$
(19)

4. Performance Evaluation

4.1. Experimental Setup

4.1.1. Dataset Description

Data are the core of artificial intelligence research, and labelling data is sometimes more important than algorithms. Existing remote sensing image datasets are private datasets. Because of sensitive data or copyright issues, many private ship datasets are difficult to disclose. To promote ship detection research, this paper proposes and establishes a set of standard, infrared remote sensing image target-ship datasets. The data are mainly derived from the image data collected by a satellite, which has an important experimental reference value. Ships in the dataset have been marked in the form of rectangular frames. In different scenarios, target ships have different sizes, directions and interferences. To verify the anti-interference ability of the algorithm, the dataset should include samples of various scenarios, such as cloud interference samples, trail interference samples, reef interference

samples, and sea-clutter interference samples to make the dataset persuasive. The images employed in the experiment include ocean scenes and nearshore scenes, which have not only calm and undisturbed sea state backgrounds but also complex backgrounds, such as clouds and reefs. The ship length is various and the ship azimuth angle is arbitrary, which is suitable for the comprehensive testing and comparison of the algorithm performances. There are 214 images in the test set, including 1270 ships. This paper enlarges the dataset by following methods: Random clipping, where a fixed size image block is intercepted randomly from the original image, and the ships in the image block have been marked in the form of a rectangular box; Mirror flip, through horizontal flip and vertical flip to construct a new dataset; Rotation transformation, where the whole image rotates in a specific direction around the fixed point; Enhanced contrast, where by changing the image's gray value to improve the visual effect of the image. We divide the image into 8456 sub-images and convert these sub-images into five datasets: dataset 1 (noninterference), which includes 4129 sub-images, including 308 target ships; dataset 2 (cloud interference), which includes 2476 sub-images, including 826 target ships; dataset 3 (trial interference), which includes 1851 sub-images, including 421 target ships; dataset 4 (reef interference), which contains 1168 sub-images, including 85 target ships; and dataset 5 (cloud interference), which containes 2788 sub-images, including 428 target ships. These datasets are classified according to interference types, and interference occurs simultaneously in multiple scenarios. However, the total number of target ships is still 1270 when repeated target ships in each sub-dataset (after classification) are excluded.

In order to evaluate the stability of the dataset and its compatibility with small sample detection, we tested and verified it through cross-validation. The specific process of the cross-validation experiment is as follows: the sample is divided into 10 parts. First, K parts are randomly selected from the sample for training, and the remaining parts are tested. To ensure the accuracy of this experiment, it was repeated three times, and then the average value of the three results was taken as the accuracy of this verification. Finally, the test accuracy of different proportions of training samples from the entire sample set is obtained, which can be used to evaluate the accuracy of the algorithm more accurately. The accuracy of the training samples is stable, which proves that the proposed method has good target expression characteristics and can effectively distinguish the target and background. At the same time, when the size of the training set exceeds 50% of the total number of samples, the test accuracy is basically stable at over 98%.

4.1.2. Applicable Platform

As the geometric progression of remote sensing image data increases and the complexity of intelligent processing algorithms increases, it is more and more difficult to process remote sensing images in real-time on the satellite platform with strictly limited resources. For large-scale remote sensing images, spaceborne resources have high requirements on algorithm running time, storage space resources and detection performance. At present, the cooperative realization of a space-borne infrared ship detection system based on DSP and FPGA has become the mainstream, but it faces two challenges: (a) it is limited by the complex space environment and hardware platform in terms of volume, weight, power consumption, transmission bandwidth alongside other aspects; and (b) satellite processors are slow to update due to the reliability, stability and cost of the equipment, which will result in their performance generally being lower than mainstream processors.

The test and verification platform of the target-detection algorithm proposed in this paper is a Tesla k40 M GPU equipped with 64 GB memory, an Ubuntu16.04 operating system, and the MATLAB 2016 language. This test environment is mainly used to help us train

the models and verify the performance of modules. As a large number of templates need to be trained in the algorithm, this test environment is needed in order to ensure the accuracy of the training results. According to the actual demand for infrared target detection, this algorithm will eventually be carried out on the FPGA platform. For example, when we select the xc7k410t from Xilinx as the core processing module of the image processing unit, it is difficult to meet the deployment requirements of deep learning methods, mainly because the compression method of deep neural networks has higher resource requirements, especially for large networks. Moreover, it is difficult to design a state-of-the-art machine for data-flow scheduling for different layers, and there will be considerable redundancy in logical resources. Therefore, deep learning methods achieve good detection accuracy in natural image target detection, but they also have great limitations during image processing with limited resources. This is the advantage of the lighter network model designed in this paper under the condition of ensuring the detection rate.

The storage space of the network model designed in this paper is less than 25 MB, which meets the requirement of limited space-borne resources and provides a feasible scheme for real-time satellite image processing. Considering the low power consumption of FPGAs, the algorithm proposed in this paper provides a feasible solution for deploying deep learning networks on satellite-borne FPGAs with guaranteed accuracy.

4.2. Effectiveness of Our Method

4.2.1. Detection Performance Verification

In this paper, the main indicators employed in the algorithm performance verification include algorithm recall rate R, algorithm accuracy P and algorithm error rate E, in which the algorithm error rate is the sum of the error rate L and error rate F, which can effectively reflect the robustness of the algorithm. The calculation process of each indicator refers to Equations (20)–(24) as follows:

$$R = \frac{D_c}{T_s},$$
 (20)

$$P = \frac{D_c}{D_s},$$
 (21)

$$F = \frac{D_f}{T_s},$$
(22)

$$L = \frac{D_l}{D_s},$$
(23)

$$\mathbf{E} = \mathbf{F} + \mathbf{L},\tag{24}$$

where D_c represents the number of correctly detected target ships and D_f represents the number of falsely detected target ships. The term D_l represents the number of missed detected target ships, D_s represents the detected targets by different methods, T_s represents the total real targets contained in the dataset, and the value of T_s is 1270. This paper compares qualitative and quantitative methods with several other representative target detection methods in the dataset. The algorithm performance was verified by datasets under different interference scenarios, as shown in Table 2.

The test results show that the proposed algorithm can achieve better detection performance in the non-interference scenario, with a recall rate of 98.2%, an accuracy of 96.8% and an error rate of only 5%. It is not difficult to determine that island interference has a great impact on the algorithm in several test scenarios. The most notable reason is that the similarity among islands and ships is high, and it is difficult to distinguish between them. The test results show that the recall rate is 92.9%, that the accuracy rate is 90.9%, and that the error rate is 16.2%, which can still reach the expected performance, reflecting that the algorithm has a strong anti-interference ability.

After multi-scene verification of our algorithm, we compare five target-detection algorithms, mainly including SVDNet [39], Faster R-CNN [40], SPP-PCANet [41], RB [42], MRA [43], and DF [44]. Among them, SVDNet is designed based on the recent popular convolutional neural networks and the singular value decompensation algorithm, it provides a simple but efficient way to adaptively learn features from remote sensing images; Faster R-CNN proposes a Region Proposal Network(RPN) that shares full-image convolutional features with the detection network, thus enabling nearly cost-free region proposals. SPP-PCANet proposes coarse-to-fine ship detection strategies based on anomaly detection and spatial pyramid pooling; RB proposes a nearly closed-form ship-rotated bounding box space used for ship detection, and designs a method to generate a small number of high-potential candidates based on this space; MRA proposes a method to densely divide a test infrared image into a set of image patches and the radiation anomaly of each patch is estimated by a Gaussian Mixture Model, thereby target candidates are obtained from anomaly image patches, then target candidates are further checked by a more discriminative criterion to obtain the final detection result; DF consists of a simple region proposal network and a deep forest ensemble, among which the region proposal network, that is trained over gradient features robustly generates a small number of candidates that precisely cover target ships in various backgrounds, and the deep forest ensemble adaptively learns features from remote sensing data and discriminates real ships from region proposals efficiently.

	None	Cloud	Trail	Reef	Clutter	Total
Ts	488	1056	658	186	524	1270
D_s	495	1095	676	189	531	1298
D _c	479	999	607	165	468	1180
D_{f}	9	57	51	21	56	116
D_1	16	96	69	24	63	92
R (%)	98.2	94.6	92.3	88.6	89.3	92.9
P (%)	96.8	91.2	89.8	87.2	88.1	90.9
E (%)	5	14.2	17.9	24.2	22.6	16.2

Table 2. Detection results of our method under different interferences.

Through the comparative analysis of the compared methods regarding their recall, precision, running time, etc. We summarize the experimental results from three parts. Compared with traditional detection methods such as RB and MRA, our method has a greatly improved detection performance and the processing unit is smoother and more efficient; compared with deep learning methods such as SVD, Faster R-CNN, SPP-PCANET and DF, our method is lighter on the premise of ensuring the detection accuracy, especially compared to the lightweight networks SVD and SP-PCANET, and shows substantial results in in terms of processing speed. In addition, there are also some space-oriented methods in the above algorithms such as SVD and MRA, despite this our method is better in terms of detection performance and processing speed, although MRA algorithms may have fewer resources because of their lack of networks. The performance comparison results are shown in Table 3. In brief, our approach is more effective at ship detection than the other three methods and takes less time to process an image.

The detection results of different methods in different scenarios are shown in Figure 8. The first line indicates thin cloud interference and has four real target ships, the second line indicates the cloud interference and sea-clutter interference scenes and has three real target ships, the third line indicates the reef interference and trail interference scenes and has one real target ship, and the fourth line indicates a ship-intensive scene with eleven real target ships. As shown in Figure 8, the yellow boxes represent the real detected targets in the original image, and the red boxes represent the detected results of different methods. Even though some false-alarm are generated due to all kinds of interference, our method achieves impressive detection performance on different sea surfaces and fewer misses and errors than other algorithms. It is proven that the proposed algorithm has high stability in different scenarios.

	SVD [39]	Faster R-CNN [40]	SPP-PCANet [41]	RB [42]	MRA [43]	DF [44]	Ours
Ts	1270	1270	1270	1270	1270	1270	1270
D_s	1293	1483	1319	1327	1288	1277	1298
D _c	1173	1188	1153	1133	1110	1144	1180
D_{f}	119	253	160	185	175	132	116
D_1	98	94	121	143	162	126	92
R (%)	92.4	93.6	90.8	89.2	87.4	90.1	92.9
P (%)	90.7	80.1	87.4	85.4	86.2	89.6	90.9
E (%)	16.9	26.3	21.8	25.4	26.4	20.3	16.2
Time/img	3.2	3.28	3.14	5.2	4.6	2.7	1.9
Model (M)	28	156	32	-	-	52	24

Table 3. Performance comparison results of the different algorithms.



Figure 8. Detection results for various methods. (a) Real target in the original image, (b) SVD, (c) Faster R-CNN, (d) SPP-PCANet, (e) RB, (f) MRA, (g) DF, and (h) our method.

4.2.2. Robustness Verification

This algorithm is a strong, robust and effective algorithm. Specifically, robustness is mainly reflected in the following three points. Firstly, the model has high accuracy or effectiveness. Secondly, small deviations from model assumptions can only have small impacts on the algorithm's performance. Thirdly, large deviations from model assumptions should not have a "catastrophic" impact on algorithm performance. So we verify function 1 with the F-measure, we verify function 2 with the True–False-positives rate, and we verify function 3 with Mean error rate.

F-measure score

The precision and recall rates are sometimes contradictory, so they need to be considered comprehensively. The F-measure is the weighted average of the precision rate and recall rate, and the F-measure value is the arithmetic mean divided by the geometric mean. When the F-measure value is small, true positives increase and false-positives decrease. Therefore, we can verify the performance of the algorithm through the F-measure score that corresponds to the first of the three function-points of robustness verification above. The calculation method for the F-measure is as follows:

$$F_{\text{score}} = \frac{(\alpha^2 + 1)P * R}{\alpha^2 (P + R)}$$
(25)

where P represents algorithm accuracy, R represents algorithm recall rate, and α represents the calculation parameter and is usually set to 1.

The F-measure weighs both precision and recall, F-score comparison results of the different algorithms are shown in Table 4, where F_{none} represents an F-score without any interference, F_{cloud} represents an F-score under cloud interference, F_{trail} represents an F-score under trail interference, F_{reef} represents an F-score under reef interference, $F_{clutter}$ represents an F-score under clutter interference, and F_{total} represents the total F-Score of detection.

	SVD [39]	Faster R-CNN [40]	SPP-PCANet [41]	RB [42]	MRA [43]	DF [44]	Ours
Fnone	0.962	0.909	0.964	0.938	0.926	0.934	0.975
F _{cloud}	0.924	0.828	0.912	0.898	0.893	0.908	0.929
F _{trail}	0.901	0.826	0.884	0.846	0.862	0.886	0.910
Freef	0.876	0.806	0.896	0.892	0.874	0.868	0.879
F _{clutter}	0.866	0.822	0.862	0.876	0.764	0.796	0.887
F _{total}	0.915	0.863	0.891	0.873	0.868	0.899	0.919

Table 4. F-score comparison results of the different algorithms.

By comparing the experimental results in Algorithm 6, it is not difficult to find that our algorithm can achieve a better F-score in different scenes, especially in the scene without interference; the F-score of the proposed method reaches 0.975. Moreover, by comparing experiments in full scenarios, it can be found that our method can also achieve better results, which proves the effectiveness and stability of the algorithm in different scenes.

True–False-positives graph

After the proposal of a region, the algorithm preliminarily obtains the potential target region, removes some background interference and negative targets, and obtains the suspected positive targets. Then, real target ships, i.e., true positive targets, are screened out through region classification, and false-alarm in the suspected positive targets, i.e., false-positive targets are eliminated. This part quantifies the performance of the algorithm under different interference scenes by using the true–false-positives graph. Because different scenes have different effects on the algorithm, the curve can intuitively reflect the adaptability of the algorithm to each scene, that is, whether there will be obvious differences in the algorithm's performance when the interference changes. Therefore, this part of the test corresponds to the second of the three function-points of the above robustness verification. The calculation of the true positives rate is as follows:

$$T_{\text{rate}} = \frac{T_P}{T_P + F_P} \tag{26}$$

where T_{rate} means the true positives rate, T_P means the true positives samples, and F_P means the false-positives samples.

In a supplemental test, we compare the correlations between the false-positive and true positive rates of different algorithms under different interferences, as shown in Figure 9.

These four scenarios are a cloud scenario, a trail scenario, a reef scenario and a seaclutter scenario. Through the linear comparison of the data in the figure, our method can quickly reach the expected detection rate when generating negative samples, which indicates that the method has good robustness and anti-interference ability. The true–



false-positive quantitative data under all scenarios are shown in Table 5, where F_P means false-positives, and T_{rate} means the true positives rate.

Figure 9. Diagram of false-positive and true-positive rates: (**a**) cloud interference; (**b**) trail interference; (**c**) reef interference; and (**d**) sea-clutter interference.

	SVD [39]	Faster R-CNN [40]	SPP-PCANet [41]	RB [42]	MRA [43]	DF [44]	Ours
$\begin{array}{c} T_{rate} \\ (F_P = 20) \end{array}$	0.468	0.436	0.475	0.448	0.456	0.418	0.568
$\begin{array}{c} T_{rate} \\ (F_P = 40) \end{array}$	0.577	0.528	0.572	0.548	0.558	0.550	0.728
$\begin{array}{c} T_{rate} \\ (F_P = 60) \end{array}$	0.645	0.596	0.656	0.650	0.634	0.662	0.776
$\begin{array}{c} T_{rate} \\ (F_P = 80) \end{array}$	0.701	0.642	0.688	0.685	0.698	0.714	0.799
T_{rate} (F _P = 100)	0.742	0.688	0.724	0.734	0.738	0.726	0.824
T_{rate} (F _P = 120)	0.768	0.712	0.761	0.755	0.756	0.752	0.838
T_{rate} (F _P = 140)	0.789	0.736	0.788	0.766	0.776	0.756	0.852
T_{rate} (F _P = 160)	0.819	0.768	0.791	0.788	0.786	0.772	0.856
T_{rate} (F _P = 180)	0.838	0.783	0.809	0.796	0.789	0.776	0.866

Table 5. True–False-positives quantitative data under all scenarios.

	SVD [39]	Faster R-CNN [40]	SPP-PCANet [41]	RB [42]	MRA [43]	DF [44]	Ours
$\begin{array}{c} T_{rate} \\ (F_{P} = 200) \end{array}$	0.849	0.796	0.816	0.811	0.808	0.798	0.867
$\begin{array}{c} T_{rate} \\ (F_P = 220) \end{array}$	0.856	0.804	0.821	0.826	0.812	0.809	0.869
$\begin{array}{c} T_{rate} \\ (F_P = 240) \end{array}$	0.868	0.807	0.836	0.841	0.827	0.814	0.875
$\begin{array}{c} T_{rate} \\ (F_P = 260) \end{array}$	0.869	0.811	0.842	0.850	0.829	0.825	0.878
$\begin{array}{c} T_{rate} \\ (F_P = 280) \end{array}$	0.873	0.813	0.844	0.855	0.842	0.831	0.882
$\begin{array}{c} T_{rate} \\ (F_P = 300) \end{array}$	0.876	0.815	0.849	0.860	0.838	0.836	0.883

Table 5. Cont.

Mean error rate

The robustness of the algorithm should be compared regarding not only the detection performance but also the avoidance of false and missed detections. To better prove the robustness of the algorithm, we compare the mean error rate obtained with different image quantities, as shown in Table 6. The third function-point of robustness is to verify the impact of large deviations on the algorithm. Since the average value is considered to be a positive correlation factor of an algorithm, the mean error rate is more meaningful than any robust measure. Therefore, this part counts the average error rate in the overall operation cycle of each algorithm, and the results can reflect the stability of the algorithm during operation, The lower the mean error rate, the better the compatibility of the algorithm with strong anti-interference.

	SVD [39]	Faster R-CNN [40]	SPP-PCANet [41]	RB [42]	MRA [43]	DF [44]	Ours
Images	214	214	214	214	214	214	214
False detection	119	253	160	185	175	132	116
Missed detection	98	94	121	143	162	126	92
Error detection	217	347	281	328	337	258	208
False/image	0.56	1.18	0.74	0.76	0.82	0.62	0.54
Missed/image	e 0.46	0.44	0.57	0.67	0.75	0.59	0.43
Error/image	1.02	1.62	1.31	1.53	1.57	1.21	0.97

Table 6. Mean error rate comparison results of the different algorithms.

5. Conclusions and Future Work

This paper presents an infrared ship detection method based on the combination of traditional feature recognition and lightweight CNN classification. The effective ship candidate-region is extracted by a multiscale feature extraction model, and the global features extracted by the Fourier transform are combined with the local features extracted by a lightweight CNN to eliminate false-alarm to confirm the target ship. Compared with the existing methods, the proposed method is more efficient and robust for target detection in complex scenes. Our future work will focus on memory storage and explore hard negative mining strategies to improve the generalisation performance.

Author Contributions: Conceptualization, R.M.; Methodology, R.M.; Validation, R.M.; Formal Analysis, R.M. and F.T.; Writing—Original Draft Preparation, R.M. and H.J.; Writing—Review and Editing, R.M., H.J. and F.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Aerospace Science and Technology Innovation Foundation of Shanghai (Grant no. SAST2020-068).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author. The data are not publicly available due to the sensitive issues.

Acknowledgments: The authors thank the assistant editor and the anonymous commentators for their insightful comments and suggestions. These suggestions have greatly improved the quality of this article.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Yang, X.; Sun, H.; Sun, H.; Yan, M.; Guo, Z.; Fu, K. Position Detection and Direction Prediction for Arbitrary-Oriented Ships via Multiscale Rotation Region Convolutional Neural Network. *IEEE Access* 2018, *6*, 50839–50849. [CrossRef]
- Zhang, Y.; Guo, L.; Wang, Z.; Yu, Y.; Liu, X.; Xu, F. Intelligent Ship Detection in Remote Sensing Images Based on Multi-Layer Convolutional Feature Fusion. *Remote Sens.* 2020, 12, 3316. [CrossRef]
- Chen, F.; Jing, Z.; Bian, M.; Zhang, Q. A Decision Mixture Model-Based Method for Inshore Ship Detection Using High-Resolution Remote Sensing Images. Sensors 2017, 17, 1470.
- 4. Tang, J.; Deng, C.; Huang, G.-B.; Zhao, B. Compressed-domain ship detection on spaceborne optical image using deep neural network and extreme learning machine. *IEEE Trans. Geosci. Remote Sens.* **2015**, *53*, 1174–1185. [CrossRef]
- 5. Yang, F.; Xu, Q.; Li, B. Ship detection from optical satellite images based on saliency segmentation and structure-LBP feature. *IEEE Geosci. Remote Sens. Lett.* 2017, 14, 602–606. [CrossRef]
- 6. Yuan, Y.; Jiang, Z.; Zhang, H.; Zhao, D.; Cai, B. Ship detection in optical remote sensing images based on deep convolutional neural networks. *J. Appl. Remote Sens.* **2017**, *11*, 042611.
- Hosang, J.; Benenson, R.; Dollár, P.; Schiele, B. What makes for effective detection proposals? *IEEE Trans. Pattern Anal. Mach. Intell.* 2015, 38, 814. [CrossRef]
- Ao, W.; Xu, F.; Li, Y.; Wang, H. Detection and Discrimination of Ship Targets in Complex Background from Spaceborne ALOS-2 SAR Images. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* 2018, 11, 536–550. [CrossRef]
- 9. Chen, C.L.P.; Li, H.; Wei, Y.; Xia, T.; Tang, Y.Y. A Local Contrast Method for Small Infrared Target Detection. *IEEE Trans. Geosci. Remote Sens.* 2014, 52, 574–581. [CrossRef]
- Deng, H.; Sun, X.; Liu, M.; Ye, C.; Zhou, X. Small Infrared Target Detection Based on Weighted Local Difference Measure. *IEEE Trans. Geosci. Remote Sens.* 2016, 54, 4204–4214. [CrossRef]
- 11. He, H.; Lin, Y.; Chen, F.; Tai, H.M.; Yin, Z. Inshore Ship Detection in Remote Sensing Images via Weighted Pose Voting. *IEEE Trans. Geosci. Remote Sens.* 2017, 55, 3091–3107. [CrossRef]
- Brauchle, J.; Bayer, S.; Berger, R. Automatic Ship Detection on Multispectral and Thermal Infrared Aerial Images Using MACS-Mar Remote Sensing Platform. In Pacific-Rim Symposium on Image & Video Technology; Springer: Cham, Switzerland, 2017.
- 13. Deng, H.; Sun, X.; Liu, M.; Ye, C. Infrared small-target detection using multiscale gray difference weighted image entropy. *IEEE Trans. Aerosp. Electron. Syst.* 2016, 52, 60–72. [CrossRef]
- 14. Lee, H.J.; Huang, L.F.; Chen, Z. Multi-frame ship detection and tracking in an infrared image sequence. *Pattern Recognit.* **1990**, *23*, 785–798. [CrossRef]
- Lebona, B.; Kleynhans, W.; Celik, T.; Mdakane, L. Ship detection using VIIRS sensor specific data. In Proceedings of the 2016 IEEE International Geoscience and Remote Sensing Symposium (IGARSS), Beijing, China, 10–15 July 2016.
- 16. Desolneux, A.; Moisan, L.; Morel, J.M. Edge Detection by Helmholtz Principle. J. Math. Imaging Vis. 2001, 14, 271–284. [CrossRef]
- 17. Faghih, F.; Smith, M. Combining spatial and scale-space techniques for edge detection to provide a spatially adaptive waveletbased noise filtering algorithm. *IEEE Trans. Image Processing* **2002**, *11*, 1062–1071. [CrossRef]
- Viola, P.A.; Jones, M. Rapid Object Detection using a Boosted Cascade of Simple Features. In Proceedings of the 2001 IEEE Computer Society Conference on Computer Vision and Pattern Recognition, Kauai, HI, USA, 8–14 December 2001.
- 19. Liu, W.; Anguelov, D.; Erhan, D.; Szegedy, C.; Reed, S.; Fu, C.Y.; Cberg, A. SSD: Single Shot MultiBox Detector. In *European Conference on Computer Vision*; Springer: Cham, Switzerland, 2016; pp. 21–37.
- 20. Fu, C.; Liu, W.; Ranga, A.; Tyagi, A.; Berg, A.C. DSSD: Deconvolutional Single Shot Detector. arXiv 2017, arXiv:1701.06659.
- Li, X.; Zhao, H.; Zhang, L. Recurrent RetinaNet: A Video Object Detection Model Based on Focal Loss. In International Conference on Neural Information Processing; Springer: Cham, Switzerland, 2018.
- 22. Zhang, S.; Wen, L.; Bian, X.; Lei, Z.; Li, S.Z. Single-Shot Refinement Neural Network for Object Detection. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, HI, USA, 21–26 July 2017.
- 23. Redmon, J.; Farhadi, A. YOLOv3: An Incremental Improvement. arXiv 2018, arXiv:1804.02767.
- 24. Chen, X.; Gupta, A. An Implementation of Faster RCNN with Study for Region Sampling. arXiv 2017, arXiv:1702.02138.

- Zhang, H.; Kyaw, Z.; Yu, J.; Chang, S.F. PPR-FCN: Weakly Supervised Visual Relation Detection via Parallel Pairwise R-FCN. In Proceedings of the IEEE International Conference on Computer Vision, Washington, DC, USA, 22–29 October 2017.
- Lin, T.; Dollar, P.; Girshick, R.; He, K.; Hariharan, B.; Belongie, S. Feature Pyramid Networks for Object Detection. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, HI, USA, 21–26 July 2017; pp. 936–944.
- Cai, Z.; Vasconcelos, N. Cascade R-CNN: Delving into High Quality Object Detection. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Salt Lake City, UT, USA, 18–23 June 2018.
- Singh, B.; Davis, L.S. An Analysis of Scale Invariance in Object Detection-SNIP. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Salt Lake City, UT, USA, 18–23 June 2018; pp. 3578–3587.
- Law, H.; Jia, D. CornerNet: Detecting Objects as Paired Keypoints. In Proceedings of the European Conference on Computer Vision (ECCV), Munich, Germany, 8–14 September 2018; pp. 1–15.
- Duan, K.; Bai, S.; Xie, L.; Qi, H.; Huang, Q.; Tian, Q. CenterNet: Keypoint Triplets for Object Detection. In Proceedings of the IEEE/CVF International Conference on Computer Vision, Seoul, Korea, 27–28 October 2019.
- 31. Law, H.; Teng, Y.; Russakovsky, O.; Deng, J. CornerNet-Lite: Efficient Keypoint Based Object Detection. In Proceedings of the 2019 IEEE/CVF Conference on Computer Vision and Pattern Recognition (CVPR), Long Beach, CA, USA, 15–20 June 2019.
- 32. Zhu, C.; He, Y.; Savvides, M. Feature Selective Anchor-Free Module for Single-Shot Object Detection. In Proceedings of the 2019 IEEE/CVF Conference on Computer Vision and Pattern Recognition (CVPR), Long Beach, CA, USA, 15–20 June 2019.
- Wang, N.; Gao, Y.; Chen, H.; Wang, P.; Zhang, Y. NAS-FCOS: Fast Neural Architecture Search for Object Detection. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, Seattle, WA, USA, 13–19 June 2020.
- Kong, T.; Sun, F.; Liu, H.; Jiang, Y.; Shi, J. FoveaBox: Beyond Anchor-based Object Detector. *IEEE Trans. Image Processing* 2019, 29, 7389–7398. [CrossRef]
- 35. Noskov, M.V.; Tutatchikov, V.S. Modification of a two-dimensional fast Fourier transform algorithm with an analog of the Cooley–Tukey algorithm for image processing. *Pattern Recognit. Image Analys.* **2017**, *27*, 110–113. [CrossRef]
- 36. Akopian, D. Fast FFT based GPS satellite acquisition methods. IEE Proc.-Radar Sonar Navig. 2017, 152, 277–286. [CrossRef]
- Zhang, Y.; Li, B.; Lu, H.; Irie, A.; Ruan, X. Sample-Specific SVM Learning for Person Re-identification. In Proceedings of the 2016 IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Las Vegas, NV, USA, 26 June–1 July 2016; pp. 1278–1287.
- Bodla, N.; Singh, B.; Chellappa, R. Soft-NMS—Improving Object Detection with One Line of Code. In Proceedings of the 2017 IEEE International Conference on Computer Vision (ICCV), Venice, Italy, 25 December 2017; pp. 2380–7504.
- 39. Zou, Z.; Shi, Z. Ship detection in spaceborne optical image with SVD networks. *IEEE Trans. Geosci. Remote Sens.* 2016, 54, 5832–5845. [CrossRef]
- Ren, S.; He, K.; Girshick, R.; Sun, J. Faster R-CNN: Towards real-time object detection with region proposal networks. In Proceedings of the International Conference on Neural Information Processing Systems, Montreal, QC, Canada, 7–12 December 2015; pp. 91–99.
- 41. Wang, N.; Li, B.; Xu, Q.Z.; Wang, Y.H. Automatic ship detection in optical remote sensing images based on anomaly detection and spp-pcanet. *Remote Sens.* **2018**, *11*, 47. [CrossRef]
- 42. Liu, Z.; Wang, H.; Weng, H.; Yang, L. Ship rotated bounding box space for ship extraction from high-resolution optical satellite images with complex backgrounds. *IEEE Geosci. Remote Sens. Lett.* **2016**, *13*, 1074–1078. [CrossRef]
- Wang, H.; Zou, Z.; Shi, Z.; Li, B. Detecting ship targets in spaceborne infrared image based on modeling radiation anomalies. *Inform. Phys. Technol.* 2017, 85, 141–146. [CrossRef]
- 44. Yang, F.; Xu, Q.; Li, B.; Ji, Y. Ship Detection from Thermal Remote Sensing Imagery through Region-Based Deep Forest. *IEEE Geosci. Remote Sens. Lett.* **2018**, *15*, 449–453. [CrossRef]